Author’s response to reviews

Title: Accuracy of parents in measuring body temperature with a tympanic thermometer designed for home use

Authors:

Joan L Robinson (jr3@ualberta.ca)
Hsing Jou (hsingjou@yahoo.com)
Donald Spady (dspady@ualberta.ca)

Version: 3 Date: 10 December 2004

Author’s response to reviews:

The reviewer correctly pointed out the fact that it is not clear that the thermometer we used as a reference standard was designed specifically for hospital use. We re-worded the manuscript to reflect the fact that we chose this model as it is one of the ones commonly used in hospitals, and would use similar technology to the tympanic thermometers used in almost all previous studies.